Author's response to reviews

Title: Phase I study of intermittent and chronomodulated oral therapy with capecitabine in patients with advanced and/or metastatic cancer

Authors:

Daniele Santini (d.santini@unicampus.it)
Bruno Vincenzi (b.vincenzi@unicampus.it)
Gaia Schiavon (g.schiavon@unicampus.it)
Annalisa La Cesa (a.lacesa@unicampus.it)
Simona Gasparro (s.gasparro@unicampus.it)
Angelo Vincenzi (aavincenzi@tiscali.it)
Giuseppe Tonini (g.tonini@unicampus.it)

Version: 4 Date: 31 January 2006

Author's response to reviews: see over
Dear Editor,

Please find enclosed the last revised version of the Original Article “Phase I study of intermittent and chronomodulated oral therapy with capecitabine in patients with advanced and/or metastatic cancer” signed by Daniele Santini, Bruno Vincenzi, Gaia Schiavon, Annalisa La Cesa, Simona Gasparro, Angelo Vincenzi and Giuseppe Tonini. As requested we have followed the editor’s and reviewers’ suggestions. Moreover, we have gone through the manuscript formatting checklist one more time and ensured that our revised manuscript conforms to all of the points. In conclusion, we made the styling and editorial changes requested by the editor.

We thank you very much for the reviewer’s criticisms that were very useful in improving our manuscript.

Awaiting your final decision,

Sincerely yours,

Daniele Santini, MD, PhD

All correspondence should be addressed to:
Daniele Santini, MD, PhD
Università Campus Bio-Medico,
Via E. Longoni n° 83, 00155 Rome, ITALY
Phone: +39-06-22541737
Fax: +39-06-22541520
E-mail: d.santini@unicampus.it